Author's response to reviews

Title: Key challenges in simulated patient programs: An international comparative case study

Authors:

Debra Nestel (debra.nestel@monash.edu)
Diana Tabak (D.Tabak@utoronto.ca)
Tanya Tierney (t.tierney@imperial.ac.uk)
Carine Layat-Burn (clayat@hecvsante.ch)
Anja Robb (anja.robb@utoronto.ca)
Susan Clark (s.clark@imperial.ac.uk)
Tracy Morrison (tracy.morrison@monash.edu)
Norma Jones (normajones1@btinternet.com)
Rachel Ellis (R.Ellis@utoronto.ca)
Cathy Smith (smith.cathy06@gmail.com)
Nancy McNaughton (N.McNaughton@utoronto.ca)
Kerry Knickle (kerry.knickle@utoronto.ca)
Jenny Higham (j.higham@imperial.ac.uk)
Roger Kneebone (r.kneebone@imperial.ac.uk)

Version: 3 Date: 5 September 2011

Author's response to reviews:

PLEASE SEE RESPONSES TO REQUESTED CHANGES (IN CAPITALS)

Formatting changes to be made:
---------------------------------
Please note responses to requested revisions:

Major revisions (we require the author to make these changes)

Abstract: this abstract does not have the correct structure. Please reformat it to use these headings: Background, Methods, Results, Conclusion. COMPLETED

Authors’ contributions: Please use initials instead of full name. COMPLETED

Tables: you have uploaded the tables as figure files. Please remove them from the submission system and include the tables within the text file of the manuscript following the references. The tables should not be embedded images or drawings and should be formatted from cells, not tabbed text. Please also move the table title to above the table and the legend to below the table, within the text COMPLETED HOWEVER FOR TABLE 2 I HAVE BEEN UNABLE TO PUT IN LANDSCAPE WITHIN THE MANUSCRIPT. IT IS PRESENT BUT IN
PORTRAIT. SORRY I HAVE BEEN UNABLE TO DO THIS FORMATTING.
TITLES ADJUSTED AS REQUESTED.

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Highlighting/tracking: Please remove from the manuscript. REMOVED.

Structure: Change introduction to background. Change author declaration to competing interests. CHANGED

Author list: Please remove titles. I WAS NOT SURE WHAT TITLES YOU MEANT. I COULD NOT SEE WHERE WE USED TITLES.

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance. COMPLETED

THANKS FOR YOUR EXCELLENT HELP.